\begin{table} %t3
\caption{\label{table:sourcelist}Source list.} \vspace*{3mm}
\par
%%\centering

\par
 \begin{tabular}{ccccccccccc}
 \hline\hline\noalign{\smallskip}
 Source\tablefootmark{a}  & {Name\tablefootmark{b}} & {RA} & {Dec} & {ID\tablefootmark{d}} &  {Magnitude} & {Redshift} & {Radio}& {$\gamma$}  &{EGRET\tablefootmark{g}} & {LAT\tablefootmark{h}} \\
{} & {} & {(J2000.0)\tablefootmark{c}} & {(J2000.0)\tablefootmark{c}} & {} & {$V$\tablefootmark{e}} & {$z$\tablefootmark{f}} & {Sample} & {Sample} &{} & {}\\
\hline\noalign{\smallskip}
0047$-$579&                         & $00^{\rm h}49^{\rm m}59\fs4731$ & $-57^\circ 38^\prime 27\farcs339$ & Q & 18.50 & 1.797$^{1}$   & Y & N & N & N\\
0208$-$512&                         & $02^{\rm h}10^{\rm m}46\fs2004$ & $-51^\circ01^\prime02\farcs891$ & B & 16.93 & 0.99$^{2}$      & Y & Y & Y & Y \\
0332$-$403&                         & $03^{\rm h}34^{\rm m}13\fs6544$ & $-40^\circ08^\prime25\farcs396$ & B & 18.50 & 1.445$^{3}$     & Y & N & N & Y \\
0405$-$385&                        & $04^{\rm h}06^{\rm m}59\fs0353$ & $-38^\circ26^\prime28\farcs042$ & Q & 18.00 & 1.285$^{4}$      & N & N & N & Y* \\
0438$-$436&                        & $04^{\rm h}40^{\rm m}17\fs1799$ & $-43^\circ33^\prime08\farcs602$ & Q & 18.8 & 2.863$^{5}$       & Y & N & N & N \\
0454$-$463&                         & $04^{\rm h}55^{\rm m}50\fs7724$ & $-46^\circ15^\prime58\farcs681$ & Q & 17.40 & 0.8528$^{6}$    & Y & Y & Y & N \\
0506$-$612&                         & $05^{\rm h}06^{\rm m}43\fs9887$ & $-61^\circ09^\prime40\farcs993$ & Q & 16.85 & 1.093$^{3}$     & N & Y & C & N \\
0518$-$458&    Pictor A       & $05^{\rm h}19^{\rm m}49\fs69$ & $-45^\circ46^\prime44\farcs5$ & G & 16.45 & 0.035058$^{7}$            & N & N & N & N \\ 
0521$-$365&   ESO 362-G 021 & $05^{\rm h}22^{\rm m}57\fs9846$ & $-36^\circ27^\prime30\farcs848$ & B & 14.50 & 0.055338$^{8}$        & Y & Y & C & N \\
%0527$-$359&                     & $05^{\rm h}29^{\rm m}20\fs810$ & $-35^\circ52^\prime34\farcs10$ & U & 18.29 & 4.172$^{(NED)}$    & Y & Y & ? \\
0537$-$441&                        & $05^{\rm h}38^{\rm m}50\fs3614$ & $-44^\circ05^\prime08\farcs934$ & Q & 15.50 & 0.894$^{1}$    & Y & Y & Y & Y \\
0625$-$354&     OH-342     & $06^{\rm h}27^{\rm m}06\fs72$ & $-35^\circ29^\prime15\farcs4$ & G & 16.50 & 0.054594$^{9}$             & Y & N & N & N \\
0637$-$752&                        & $06^{\rm h}35^{\rm m}46\fs5079$ & $-75^\circ16^\prime16\farcs814$ & Q & 15.75 & 0.653$^{10}$   & Y & N & N & N \\
1104$-$445&                        & $11^{\rm h}07^{\rm m}08\fs6929$ & $-44^\circ49^\prime07\farcs567$ & Q & 18.20 & 1.598$^{1}$    & Y & N & N & N \\
1144$-$379&                       & $11^{\rm h}47^{\rm m}01\fs3706$ & $-38^\circ12^\prime11\farcs022$ & Q & 16.20 & 1.048$^{11}$    & N & N & N  & Y \\
1257$-$326&                       & $13^{\rm h}00^{\rm m}42\rlap{.}^{\rm s}4259$ & $-32^\circ53^\prime12\farcs110$ & Q & 18.70 & 1.256$^{12}$                & N & N & N & N \\
1313$-$333&                       & $13^{\rm h}16^{\rm m}07\fs9859$ & $-33^\circ38^\prime59\farcs171$ & Q & 20.00 & 1.21$^{13}$     & N & Y & C & N \\
1322$-$428&   Cen A, NGC 5128  & $13^{\rm h}25^{\rm m}27\fs6152$ & $-43^\circ01^\prime08\farcs805$ & G & 7.84 & 0.001825$^{14}$     & Y & Y & Y  & Y \\
1323$-$526&   PMN J1326-5256   & $13^{\rm h}46^{\rm m}48\fs70$ & $-52^\circ56^\prime22\farcs0$ & U & \dots & \ldots                 & N & Y & C$^{30}$ & N \\
1333$-$337&  IC 4296    & $13^{\rm h}36^{\rm m}39\fs05$ & $-33^\circ57^\prime57\farcs2$ & G & 11.61 & 0.012465$^{15}$               & N & N & N & N \\
1424$-$418&             & $13^{\rm h}27^{\rm m}56\fs2975$ & $-42^\circ06^\prime19\farcs437$ & Q & 17.7 & 1.522$^{16}$               & Y & Y & Y & N \\
1454$-$354&                 & $14^{\rm h}57^{\rm m}26\fs7117$ & $-35^\circ39^\prime09\farcs971$ & Q & 19.50 & 1.424$^{17}$          & Y & Y & Y$^{31}$ & Y \\
1501$-$343& PMN J1505-3432  & $15^{\rm h}05^{\rm m}02\fs4$ & $-34^\circ32^\prime57\farcs0$ & U & \ldots &  \ldots                   & N & Y & Y$^{31}$ & N \\
1549$-$790&         & $15^{\rm h}56^{\rm m}58\fs8697$ & $-79^\circ14^\prime04\farcs281$ & G & 18.50 & 0.1501$^{18}$                 & Y & N & N & N \\
1610$-$771&               & $16^{\rm h}17^{\rm m}49\fs2726$ & $-77^\circ17^\prime18\farcs467$ & Q & 19.00 & 1.71$^{19}$             & Y & N & N & N \\
1714$-$336&             & $17^{\rm h}17^{\rm m}36\fs0293$ & $-33^\circ42^\prime08\farcs829$ & B &  & \ldots                         & N & Y & Y$^{31}$ & N \\
1716$-$771&             & $17^{\rm h}23^{\rm m}50\fs51$ & $-77^\circ13^\prime50\farcs1$ & U & \ldots & \ldots                  & N & Y & C$^{32}$ & N \\
1718$-$649& NGC 6328    & $17^{\rm h}23^{\rm m}41\fs0296$ & $-65^\circ00^\prime36\farcs615$ & G & 13.16 & 0.014428$^{20}$                & Y & N & N & N \\
1733$-$565&               & $17^{\rm h}37^{\rm m}35\fs7706$ & $-56^\circ34^\prime03\farcs155$ & G & 18.00 & $0.098^{21}$            & Y & N & N & N \\
1759$-$396&           & $18^{\rm h}02^{\rm m}42\fs680$ & $-39^\circ40^\prime07\farcs905$ & Q & ? & 0.296$^{22}$                     & N & Y & Y & Y* \\
1804$-$502& PMN J1808-5011& $18^{\rm h}08^{\rm m}13\fs90$ & $-50^\circ11^\prime54\farcs0$ & Q & \ldots & \ldots                     & N & Y & C$^{32}$ & N \\
1814$-$637&                         & $18^{\rm h}19^{\rm m}35\fs0023$ & $-63^\circ45^\prime48\farcs189$ & G & 18.00 & 0.06270$^{23}$& N & N & N & N \\
1933$-$400&              & $19^{\rm h}37^{\rm m}16\fs2173$ & $-39^\circ58^\prime01\farcs552$ & Q & 18.00 & 0.965$^{24}$             & N & Y & Y & N \\
1954$-$388&            & $19^{\rm h}57^{\rm m}59\fs8192$ & $-38^\circ45^\prime06\farcs356$ & Q & 17.70 & 0.63$^{25}$                & Y & N & N & N \\
2005$-$489&                       & $20^{\rm h}09^{\rm m}25\fs3906$ & $-48^\circ49^\prime53\farcs720$ & B & 15.30 & 0.0710$^{26}$   & N & Y & Y$^{33}$ & Y \\
2027$-$308& ESO 462-G 027                          & $20^{\rm h}30^{\rm m}57\fs934$ & $-30^\circ39^\prime24\farcs35$ & G & ? &\ldots& N & Y & Y$^{31}$ & N \\
2052$-$474&                & $20^{\rm h}56^{\rm m}16\fs3598$ & $-47^\circ14^\prime47\farcs627$ & Q & 19.10 & 1.489$^{27}$           & Y & Y & Y & Y \\
2106$-$413&                         & $21^{\rm h}09^{\rm m}33\fs1885$ & $-41^\circ10^\prime20\farcs605$ & Q & 21.00 & 1.058$^{16}$  & Y & N & N & N \\
2149$-$306&                         & $21^{\rm h}51^{\rm m}55\fs5239$ & $-30^\circ27^\prime53\farcs697$ & Q & 18.40 & 2.345$^{3}$   & N & N & N & N \\
2152$-$699&  ESO 075-G 041          & $21^{\rm h}57^{\rm m}05\fs9805$ & $-69^\circ41^\prime23\farcs685$ & G & 1430 & 0.028273$^{28}$& N & N & N & N \\
2155$-$304&                       & $21^{\rm h}58^{\rm m}52\fs0651$ & $-30^\circ13^\prime32\farcs118$ & B & 14.00 & 0.116$^{29}$    & N & Y & Y & Y \\
2204$-$540&                  & $22^{\rm h}07^{\rm m}43\fs7332$ & $-53^\circ46^\prime33\farcs820$ & Q & 18.00 & 1.206$^{3}$          & Y & N & N & Y \\
2326$-$477&              & $23^{\rm h}29^{\rm m}17\fs7043$ & $-47^\circ30^\prime19\farcs115$ & Q & 16.79 & 1.2990$^{1}$             & Y & N & N & N \\
2355$-$534&               & $23^{\rm h}57^{\rm m}53\fs2661$ & $-53^\circ11^\prime13\farcs689$ & Q & 17.80 & 1.0060$^{27}$           & N & N & N & N \\
\hline
\end{tabular}
\tablefoot{\tablefoottext{a}{ IAU source designation.} %\\
\tablefoottext{b}{ Alternative source name where appropriate.} %\\
\tablefoottext{c}{ Right ascension and declination (J2000.0).} %\\
\tablefoottext{d}{ The optical counterpart, denoted as follows: (G) galaxy,
(Q) quasar,  (B) BL~Lac object, or (U) unclassified. 
} %\\
\tablefoottext{e}{ Optical magnitude.} %\\
\tablefoottext{f}{ Redshift.} %\\
\tablefoottext{g}{Y = Detected, N = Not Detected, C = Candidate. Those detections and candidates without a reference indicated are from \citet{Hartman1999}.} %\\
\tablefoottext{h}{Based on the LAT 3-month list. ``*'' denotes a low confidence detection.}
\tablebib{(1)~\citet{Peterson1976}; 
(2)~\citet{Wisotzki2000};
(3)~\citet{Hewitt1989};
(4)~\citet{Morton1978};
(5)~\citet{Stickel1994};
(6)~\citet{Sulentic2004};
(7)~\citet{Lauberts1989};
(8)~\citet{Keel1985};
(9)~\citet{Quintana1995};
(10)~\citet{Huntstead1978};
(11)~\citet{Stickel1989};
(12)~\citet{Perlman1998};
(13)~\citet{Jauncey1982};
(14)~\citet{Graham1978};
(15)~\citet{Smith2000};
(16)~\citet{White1988};
(17)~\citet{Jackson2002};
(18)~\citet{Tadhunter2001};
(19)~\citet{Huntstead1980};
(20) HI Parkes All Sky Survey Final Catalog;
(21)~\citet{Tadhunter1993};
(22)~\citet{Liang2003};
(23)~\citet{Danziger1979b};
(24)~\citet{Drinkwater1997};
(25)~\citet{Browne1975};
(26)~\citet{Falomo1987};
(27)~\citet{Jauncey1984};
(28)~\citet{daCosta1991};
(29)~\citet{Falomo1993};
(30)~\citet{Bignall2008};
(31)~\citet{Sowards-Emmerd2004};
(32)~\citet{Tornikoski2002};
(33)~\citet{Lin1997}.
}}\vspace*{6mm}
\end{table}